Author's response to reviews

Title: Identification of Biomarkers in Ductal Carcinoma in situ of the Breast with microinvasion

Authors:

Yasuhiro Okumura (sigeoku@yahoo.co.jp)
Yutaka Yamamoto (ys-yama@triton.ocn.ne.jp)
Zhenhuan Zhang (zhangzhen@mail.nih.gov)
Tatsuya Toyama (t.toyama@med.nagoya-cu.ac.jp)
Teru Kawasoe (tkawasoe@kumamoto-u.ac.jp)
Mutsuko Ibusuki (pellegrino2002jp@yahoo.co.jp)
Yumi Honda (yumih@kumamoto-u.ac.jp)
Ken-ichi Iyama (iyama@kumamoto-u.ac.jp)
Hiroko Yamashita (hirokoy@med.nagoya-cu.ac.jp)
Hirotaka Iwase (hiwase@kumamoto-u.ac.jp)

Version: 8 Date: 2 October 2008

Author's response to reviews:

Dear Sir/Madam,

Thank you for your direction.

1. I removed the Running title and Article type from title page.
2. I changed the order of table 1,2 and cited adequately.
3. I presented Reference citations in black.
4. I renamed Introduction to Backgroung, Materials and Methods to Methods.

Sincerely yours,

Hirotaka Iwase